Author's response to reviews

Title: A spindle cell carcinoma presenting with osseous metaplasia in the gingiva: a case report with immunohistochemical analysis.

Authors:

Naoki Katase (gmd17099@cc.okayama-u.ac.jp)
Ryo Tamamura (tamamura@md.okayama-u.ac.jp)
Mehmet Gunduz (mgunduz@md.okayama-u.ac.jp)
Jun Murakami (jun-m@md.okayama-u.ac.jp)
Jun-Ichi Asaumi (asaumi@md.okayama-u.ac.jp)
Goichi Tsukamoto (gtsuka@md.okayama-u.ac.jp)
Akira Sasaki (asasaki@md.okayama-u.ac.jp)
Hitoshi Nagatsuka (jin@md.okayama-u.ac.jp)

Version: 4 Date: 28 November 2008

Author's response to reviews:

we are so sorry to still have some mistakes after the revision. We corrected the Affiliation and included Consent section between conclusion section and abbreviation section.